Author's response to reviews

Title: Serum procalcitonin for the early recognition of nosocomial infection in the critically ill patients: a preliminary report

Authors:

Pierre Emmanuel Charles (pierre-emmanuel.charles@chu-dijon.fr)
Emmanuel Kus (manu.kus@libertysurf.fr)
Serge Aho (ludwig-serge.aho@chu-dijon.fr)
Sebastien Prin (sebastien.prin@chu-dijon.fr)
Jean-Marc Doise (jean-marc.doise@chu-dijon.fr)
Nils-Olivier Olsson (nils-olivier.olsson@chu-dijon.fr)
Bernard Blettery (bernard.blettery@chu-dijon.fr)
Jean-Pierre Quenot (jean-pierre.quenot@chu-dijon.fr)

Version: 6 Date: 6 April 2009

Author's response to reviews:

Dear Editor,

as required by Reviewer 2, the word sepsis has been replaced by "infection" within the Figure 2 legend. The english expression has also been revised.

The format has also been checked.

We hope that you will find suitable for publication the newly submitted manuscript.

Best regards

PE CHARLES